Author’s response to reviews

Title: Patient education integrated with acupuncture for relief of cancer-related fatigue: Randomized controlled feasibility study

Authors:

  Michael F Johnston (johnston@ucla.edu)
  Ron D Hays (drhays@ucla.edu)
  Saskia K Subramanian (Saskia@ucla.edu)
  Robert M Elashoff (RElashoff@biomath.ucla.edu)
  Eleanor K Axe (EAxe@mednet.ucla.edu)
  Jie-Jia Li (JiLi@mednet.ucla.edu)
  Irene Kim (IKim@mednet.ucla.edu)
  Roberto B Vargas (RBVargas@mednet.ucla.edu)
  Jihey Lee (Jihey.Lee@gmail.com)
  LuGe Yang (LuGeYang@ucla.edu)
  Ka-Kit Hui (khui@mednet.ucla.edu)

Version: 6 Date: 21 May 2011

Author’s response to reviews:

Melissa Norton, MD
Editor in Chief
BMC Complementary and Alternative Medicine

May 20, 2011

Dear Dr. Norton,

Thank you for your patience with us. We have changed the abstract in the webpage so it is consistent with the abstract in the paper. In reviewing the paper, we noticed that the third page is entirely blank. We are unsure of whether you did this intentionally or not.

We are delighted that our manuscript will be published in BMC Complementary and Alternative Medicine.

Warm regards,

Ka-Kit Hui, MD, FACP
Director, Center for East-West Medicine
Wallis Annenberg Professor in Integrative East-West Medicine
Chair, Collaborative Centers of Integrative Medicine at UCLA
Department of Medicine
David Geffen School of Medicine at UCLA